Author’s response to reviews

Title: Comparison of CLART HPV2 genotyping assay to Linear Array and Hybrid Capture 2: a split-sample study

Authors:

Ditte M Ejegod (ditte.ejegod@regionh.dk)
Matejka Rebolj (Matejka.rebolj@regionh.dk)
Jesper Bonde (jesper.bonde@hyh.regionh.dk)

Version: 3
Date: 9 March 2015

Author’s response to reviews: see over
Thank you for your mail. We have elaborated on the informed consent procedure as requested. All additions to this point are marked in gray, below and in the text:

“LA and HC2 data were collected with informed consent as part of the Danish arm of a multicenter European trial (ClinicalTrial.gov Identifier: NCT01671462), approved by the Danish Capital Region Ethical Committee (H-2012-070). Informed consent was obtained by the sample taking gynecologists, and maintained in the women’s patient records, as well as in copy at the Department of Pathology in concordance with Danish Ethical guidelines. Additional testing on CLART, not used for clinical management, was undertaken as a quality development study, for which ethical approval and informed consent are not required, in concordance with the current Danish law.”

We hope this satisfies your requirements, otherwise please do not hesitate to contact me again. I look forward to a conclusion to this process and as always we appreciate publishing through BMC.

Sincerely Yours